Author's response to reviews

Title: Epigenetic regulation of lncRNA connects ubiquitin-proteasome system with infection-inflammation in preterm births and preterm premature rupture of membranes

Authors:

Xiucui Luo (lxc8816@sina.com)
Jing Pan (lygkjk@126.com)
Leilei Wang (wangleileiok@hotmail.com)
Peirong Wang (wangpeirong@gmail.com)
Meijiao Zhang (940651562@qq.com)
Meilin Liu (591453886@qq.com)
Ziqing Dong (gingqing-dzq@126.com)
Qian Meng (lygmg6326@163.com)
Xuguang Tao (xgtltw2005@gmail.com)
Xinliang Zhao (xinliang.zhao@gmail.com)
Julia Zhong (Julia.zhong@gmail.com)
Weina Ju (weina.ju@yahoo.com)
Yang Gu (815501606@qq.com)
Edmund C. Jenkins (ed.jenkins@opwdd.ny.gov)
Ted Brown (Ted.brown@opwdd.ny.gov)
Qingxi Shi (shiqingxi0220@163.com)
Nanbert Zhong (nanbert.zhong@opwdd.ny.gov)

Version: 5
Date: 21 January 2015

Author's response to reviews:

Responding to the reviewer’s comments, we have revised our manuscript as recommended in the following:

1, Data Deposition:
A: Our data generated by microarray has been deposited in Gene Expression Omnibus with an accession number GSE 50879 (http://www.ncbi.nlm.nih.gov/geo/info/linking.html). This sentence is included in page 16, lines 270-272.

2, Availability of Supporting Data:
A: There is no other “Supporting Data” for this manuscript.

3, Please include the full name of the hospital ethics committee which approved your study in the methods section.
A: The Ethics Committee of Lianyungang Maternal and Children's Hospital reviewed and approved the research project (Please see page 11, line 173).
Thanks,
Nanbert Zhong,